Author’s response to reviews

Title: Self-motivated medical care-seeking behaviors and disease progression in a community-based cohort of chronic hepatitis B virus-infected patients in China

Authors:
Hui Zheng (zhenghui@chinacdc.cn)
Guomin Zhang (zhanggm@chinacdc.cn)
Fuzhen Wang (wangfz@chinacdc.cn)
Zundong Yin (yinzd@chinacdc.cn)
Miao Ning (miaoning@chinacdc.cn)
Xiaojin Sun (sunxj@chinacdc.cn)
Xiaofeng Liang (laingxf@chinacdc.cn)
Fuqiang Cui (nip2018@163.com)

Version: 5 Date: 27 May 2019

Author’s response to reviews:

1. Map Attribution
-- Please state in the cover letter whether the map depicted in figure 1B is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

Should you require an alternative source you may wish to try Wikimedia Commons:
http://commons.wikimedia.org/wiki/Main_Page

Response: done, we have added this in the last sentence of the cover letter as ‘And we declare that the map depicted in figure 1B is our own, not taking from other sources.’

2. Funding
-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: done. we have added this information in Funding section as ‘The funding body was not involved in the design of the study or in collection or analysis of the data.’

3. Written Consent

-- Please can you clearly state under ethics approval that consent was written.

Response: done, we have changed the Ethics approval and consent to participate section as following:

The ethics approval for this study was signed by the China CDC Ethical Review Committee (Ethics certificate:201339), and the study was conducted following the ethics regulations. Cohort participants were informed of the study purpose and their right to keep information confidential. Before the investigation, an informed consent form was signed by all participants.

4. Authors and Contributions

-- Please can you clearly state all authors read AND approved the final version of the manuscript.

Response: done, we have changed this section as following:

Conceived and designed the study: LXF, CFQ. Collected the data: ZH, ZGM, WFZ, YZD, MN, SXJ. Analyzed and interpreted the data: ZH, ZGM, WFZ. Wrote the first draft of the manuscript: ZH. Supervised the study: LXF and CFQ. All authors were involved in revising the article critically for important intellectual content, and all authors had read and approved the final version of the manuscript.

5. Second Copy

-- Thank you for providing your tracked changes. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

Response: done, we had canceled the tracked document.